Author's response to reviews

Title: Modulation of macrophage cytokine profile during evolution of solid tumors: susceptibility to Candida albicans infection

Authors:

Marcela R Camargo (marcelarc@fc.unesp.br)
James Venturini (james@fc.unesp.br)
Fátima R Vilani-Moreno (fmoren@iisl.br)
Maria Sueli P Arruda (sueli@fc.unesp.br)

Version: 5 Date: 23 May 2009

Author's response to reviews:

The BioMed Central Editorial Production Team:

Major revisions (We require the author to make these changes). Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Authors:
The list of authors in the manuscript was modified according to formatting requirements.

The BioMed Central Editorial Production Team:

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Abbreviations - Please include these as a paragraph, not a list.

Authors:
The list of Abbreviations in the manuscript was modified according to formatting requirements.

The BioMed Central Editorial Production Team:

Table title and legend - The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.

Authors. The table title and legend were modified according to formatting requirements.